Author's response to reviews

Title: Innate immunity in ocular Chlamydia trachomatis infection: contribution of IL8 and CSF2 gene variants to risk of trachomatous scarring in Gambians

Authors:

   Angels Natividad (angels.natividad@inserm.fr)
   Jeremy Hull (jhull@molbiol.ox.ac.uk)
   Luoni Gaia (gaia@well.ox.ac.uk)
   Martin Holland (mholland@mrc.gm)
   Kirk Rockett (Kirk.Rockett@well.ox.ac.uk)
   Hassan Joof (hmjoof@yahoo.co.uk)
   Matthew Burton (matthew.burton@lshtm.ac.uk)
   David Mabey (david.mabey@lshtm.ac.uk)
   Dominic Kwiatkowski (dominic@well.ox.ac.uk)
   Robin Bailey (robin.bailey@lshtm.ac.uk)

Version: 4 Date: 21 November 2009

Author's response to reviews:

All major and minor formatting changes have been made to the revised article submitted here.